Author’s response to reviews

Title: Hygienic safety of reusable tap water filters (Germlyser(R)) with operating time of 4 respectively 8 weeks in a haematological oncology transplantation unit

Authors:

Georg Daeschlein (daeschlein@uni-greifswald.de)
William H Kruger (william.krueger@uni-greifswald.de)
Christian Selepko (cselepko@web.de)
Niels Rochow (rochow@uni-greifswald.de)
Gottfried Dolken (doelken@uni-greifswald.de)
Axel Kramer (kramer@uni-greifswald.de)

Version: 3 Date: 20 May 2007

Author’s response to reviews: see over
To: The BioMed Central Editorial Production Team


Betreff: Manuscript submission: ‘Hygienic safety of reusable tap water filters (Germlyser®) with operating time of 4 resp. 8 weeks in a haematological oncology transplantation unit’

Datei: BrieEMAU

To whom it may concern!

Thank you for your kind mail from the 15th of May.

We have addressed all changes required for publication as follows:

- **Title:** Has been changed to sentence case
- **Author list:** Author list in manuscript and submission system are now identical. Special german spelling (e.g. Krüger instead of Kruger) has been implemented into system
- **Abstract:** Is the same in system and manuscript. Abstract has been re-structured as desired.
- **Manuscript sections:** Manuscript has been re-structured as desired, acknowledgements are not applicable
- **Introduction:** Renamed to Background
- **Materials & Methods:** Renamed to Methods
- **Authors’ contributions:** Has been corrected and moved as desired
- **Acknowledgements:** Not applicable
- **Figure legends:** Has been moved
- **Tables and Legends to figures Header:** Removed
- **References:** Authors were inserted and the term et al. does not occur any longer. In this context I would like to notify you that the restriction auf authors was generated automatically by Reference Manager’s BMC-style.

Please notify us in the case that any further modifications are necessary for publication.

Yours, sincerely,

Priv.-Doz. Dr. W. H. Krüger
Enclosure:

- Manuscript (revised version #2, file name: H2O-BMC3.doc): ‘Hygienic safety of reusable tap water filters (Germlyser®) with operating time of 4 resp. 8 weeks in a haematological oncology transplantation unit’